Author's response to reviews

Title: Disease-specific survival for limited-stage small-cell lung cancer affected by statistical method of assessment

Authors:

Patricia Tai (patricia.tai@scf.sk.ca)
Judith-Anne W Chapman (JChapman@ctg.queensu.ca)
Edward Yu (Edward.Yu@lhsc.on.ca)
Dennie Jones (DVJones@salud.unm.edu)
Changhong Yu (CYu@ctg.queensu.ca)
Fei Yuen (Fyuan@ctg.queensu.ca)
Sang-Joon Lee (SJLee@salud.unm.edu)

Version: 6 Date: 1 August 2006

Author's response to reviews:

I changed all the "modeling" to "modelling" so that the whole manuscript is similar throughout.